Author's response to reviews

Title: Conducting Online Expert Panels: A Feasibility and Experimental Replicability Study

Authors:

Dmitry Khodyakov (dkhodyak@rand.org)
Susanne Hempel (Susanne_Hempel@rand.org)
Lisa Rubenstein (Lisa.Rubenstein@va.gov)
Paul Shekelle (Shekelle@rand.org)
Robbie Foy (R.Foy@leeds.ac.uk)
Susanne Salem-Schatz (SSS@hcqi.com)
Sean O’Neill (Sean.M.Oneill@gmail.com)
Margie Danz (MJSDanz@gmail.com)
Siddhartha Dalal (SDalal@rand.org)

Version: 4 Date: 19 December 2011

Author's response to reviews:

Hello,

I was able to make all but one requested changes. The only potentially outstanding issue is cropping of the image. I really do not understand what needs to be cropped, so I would appreciate your help with this.

Thanks,

Dmitry Khodyakov